\documentclass{article}

\usepackage{epsfig}

\pagestyle{empty}


\setlength{\topmargin}{-20pt}
%\setlength{\textwidth}{470pt}
%\setlength{\textheight}{650pt}

\begin{document}

\Large


\begin{center}
Programming in Perl \\
66-2230-01 \\
Wednesdays 5-6:50, March 18 - April 29, DCC 324
\end{center}

\vspace {0.1in}

\large
\begin{center}
\begin{tabular}{||ll||} \hline \hline
\multicolumn{2}{||c||}{Instructor (Louis Ziantz)} \\ \hline 
{\bf Office : } & Amos Eaton 427 \\ 
         & (sometimes found in AE 217) \\ 
{\bf Office Hours : } &  Tue. 12-2, Fri. 1-3 \\ 
               & and by appointment \\ 
{\bf E-mail :       } & ziantzl@cs.rpi.edu \\ 
               & (best way to contact me) \\ \hline \hline
\multicolumn{2}{||c||}{TA (Yu Jiang)} \\ \hline 
{\bf Office : } & Sage 1112 \\ 
{\bf Office Hours : } &  Mon 10-12, Wed 10-12 \\ 
               & and by appointment \\ 
{\bf Phone : }        & 276-4849 \\ 
{\bf E-mail :       } & yujiang@cs.rpi.edu \\ \hline \hline
\multicolumn{2}{||c||}{TA (Zhong Zhang)} \\ \hline 
{\bf Office : } & Amos Eaton 316W \\ 
{\bf Office Hours :  } & Tue. 2-4, Thur. 3-5 \\ 
               & and by appointment \\ 
{\bf Phone : }        & 276-8057 \\ 
{\bf E-mail :       } & zhangz@cs.rpi.edu \\ \hline \hline
\end{tabular}
\end{center}

\vspace {0.1in}

\centerline{\psfig{file=kyle2.ps,width=2in}
%%\psfig{file=kenny2.ps,width=1.25in}
%%\psfig{file=kyle2.ps,width=1.25in}
%%\psfig{file=stan2.ps,width=1.25in} 
}
%% price02c++.grey.ps
\begin{center}
%%The publishers of the ``for Dummies'' books find a new market.
%%C++ is everyone's best friend.
%%C++ is sweet!

%So my friend is taking a class in Perl.  And I was all like, ``No way
%dude!'' And he said, ``Definitely dude!''  I was like, ``Whoa dude,
%that rules!''  To which he replied, ``Sure does dude!''  ``Dude, I bet
%John Elway uses Perl all the time...''

So my friend is taking a class in Perl.  And I am all like, ``Whoa dude,
that rules!  I bet
John Elway uses Perl all the time, dude...''

\end{center}

\end{document}







